Author's response to reviews

Title: A case of a four-year-old child adopted at eight months with unusual mood patterns and significant polypharmacy.

Authors:

Magdalena Romanowicz (romanowicz.magdalena@mayo.edu)

Jennifer Vande Voort (VandeVoort.Jennifer@mayo.edu)

Alastrair McKean (mckean.alastair@mayo.edu)

Version: 1 Date: 29 Dec 2016

Author’s response to reviews:

Dear Editor,

Written informed consent for the publication of this case was obtained from the child's adoptive parent (legal guardian).

We have used a made up name for this patient to protect his identity. We have changed it now into initial('J')instead of the name 'Jack'.

Regards,

M. Romanowicz.